Author’s response to reviews

Title: Helicobacter-induced Gastric Inflammation Alters the Properties of Gastric Tissue Stem/Progenitor Cells

Authors:

Wataru Shibata (wshibata@yokohama-cu.ac.jp)
Soichiro Sue (ssue@yokohama-cu.ac.jp)
Sachiko Tsumura (e133055g@yokohama-cu.ac.jp)
Yasuaki Ishii (yasu8720@yahoo.co.jp)
Takeshi Sato (dorcus_rectus@msn.com)
Eri Kameta (lesyeuxfermes_ek@yahoo.co.jp)
Makoto Sugimori (mak.sugimori@gmail.com)
Hiroaki Yamada (hiro0118@yokohama-cu.ac.jp)
Hiroaki Kaneko (hero5722@yokohama-cu.ac.jp)
Tomohiko Sasaki (ssktmhk@yokohama-cu.ac.jp)
Tomohiro Ishii (tishii@yokohama-cu.ac.jp)
Tosshihide Tamura (t_tamura@yokohama-cu.ac.jp)
Masaaki Kondo (mkondou@yokohama-cu.ac.jp)
Shin Maeda (smaeda@yokohama-cu.ac.jp)

Version: 2 Date: 21 Nov 2017

Author’s response to reviews:

Thank you for checking our manuscript. As per your requests, we edited the manuscript as follows:

1. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different
colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

- Yes, we did.

2. Remove the response to reviewers, as it is no more necessary at this stage.

- Yes, we did.

3. Please include the full name of the ethics committee that approved the study and the committee’s reference number if appropriate.

- Yes, we did.

4. Please amend your statement in 'Consent for Publication' in Declarations, and state 'not applicable'.

- Yes, we did.

5. We noted that the author Makoto Sugimori is not cited in the Authors' Contributions section: please provide his contribution indicating him with his full initials (MS).

- We added his initials in the contribution part.

6. Please provide citation of the supplementary material after the Figure Legends (and continue to cite them within the manuscript), with the supplementary material legend, and a brief description of the data included.

- We fixed it.